Author’s response to reviews

Title: PAX6 gene variations associated with aniridia in south India

Authors:
Guruswamy Neethirajan (neethirajan@aravind.org)
Subbaiah RAMASAMY Krishnadas (krishnadas@aravind.org)
Perumalsamy Vijayalakshmi (pvijayalakshmi@aravind.org)
Shetty Shashikanth (drshashikants@yahoo.com)
Dr Periasamy Sundaresan (sundar@aravind.org)

Version: 4 Date: 14 Apr 2004

Dear Dr. Matthew Hodgkinson

Editorial Assistant, BMC Medical Genetics

We have modified the small corrections in our manuscript entitled “PAX6 gene variations associated with aniridia in south India” (1358717106269565) as per your comment.

Please intimate the date of publication before one day

We look forward to hear from you soon

Yours sincerely

P.Sundaresan

Dr.P.Sundaresan
Department of Genetics
Aravind Medical Research Foundation
No.1 Anna Nagar
Madurai-625 020
Tamilnadu, INDIA
Email:sundar@aravind.org
Phone: 91 452 2532653, 5356100
Fax: 91 452 2530984

Formatting changes requested:

Table 2 - The borders of tables included as part of the main manuscript must be visible black lines. Tables must be divided into cells / fields. Tables generated with tabbed text are not acceptable.

Changes have been made

Page 4, paragraph 2 - Revise "Eventhough PAX6 gene was studied in various ethnic populations, Indian aniridia population is not a well-studied group, and hence we studied the PAX6 variations." to:
"Even though the PAX6 gene has been studied in various ethnic populations,
Indian aniridia mutations are not well studied. For this reason we decided to study PAX6 variations in an Indian population."

Changes have been made


Changes have been made

Reference 19 - Change to: Human PAX6 Allelic Variant Database [http://pax6.hgu.mrc.ac.uk/]

Changes have been made